Author's response to reviews

Title: Fast growth associated with aberrant vasculature and hypoxia in fibroblast growth factor 8b (FGF8b) over-expressing PC-3 prostate tumour xenografts

Authors:

Johanna Tuomela (jomitu@utu.fi)
Tove J Grönroos (tovgro@utu.fi)
Maija P Valta (mapava@utu.fi)
Jouko Sandholm (jouko.sandholm@btk.fi)
Aleksi Schrey (aleksi.schrey@utu.fi)
Jani Seppänen (jani.seppanen@pharmatest.fi)
Päivi Marjamäki (paivi.marjamaki@utu.fi)
Sarita Forsback (sarita.forsback@utu.fi)
Ilpo Kinnunen (ilpo.kinnunen@utu.fi)
Olof Solin (olof.solin@utu.fi)
Heikki Minn (heikki.minn@utu.fi)
Pirkko L Härkönen (harkonen@utu.fi)

Version: 3 Date: 28 October 2010

Author's response to reviews: see over
Dear Dr Norton,

Thank you for your letter of acceptance 25th Oct 2010. Manuscript MS: 1275011302374708 is now corrected and I list here all changes:

1) I changed the format of affiliations to submission system and it’s now identical with the manuscript.

2) I added reference 46 to the manuscript page 13, last paragraph.

3) Abstract is corrected according to the suggestions of reviewer Anthony Joshua: 4th line of abstract “and being associated with poor prognosis” to "and is associated with a poor prognosis" and 13th line of abstract - grew "very fast" to "grew rapidly".

We hope that the manuscript is now acceptable.

Yours sincerely,

Johanna Tuomela, PhD

Department of Cell Biology and Anatomy, University of Turku, Kiinamyllynkatu 10, FI-20520 Turku, Finland, Tel +358-407343520, Fax +358-2-3337352, E-mail: jomitu@utu.fi